Author's response to reviews

Title: Rotavirus and Norovirus infections among acute gastroenteritis children in Morocco

Authors:

Maria El qazoui (elqazouimaria@yahoo.fr)
Hicham Oumzil (droumzil@gmail.com)
Larbi Baassi (larbi.baassi@gmail.com)
Nezha El Omari (nouzha_elomari@yahoo.fr)
Khalid Sadki (ksadki1@yahoo.fr)
Saaid Amzazi (amzazi@gmail.com)
Mohamed Benhafid (benhafidm@yahoo.fr)
Rajae El Aouad (rajaelaoud@yahoo.fr)

Version: 3
Date: 12 May 2014

Author's response to reviews: see over
Author’s response to reviews

Title: Rotavirus and Norovirus infections among acute gastroenteritis children in Morocco

Authors: Maria El Qazoui, Hicham Oumzil, Larbi Baassi, Nezha El Omari, Khalid Sadki, Saaïd Amzazi, Mohamed Benhafid, and Rajae El Aouad

Version: 2 Date: 8 May 2014

Author’s response to reviews: See over
Reviewer’s report

Title: Rotavirus and Norovirus infections among acute gastroenteritis children in Morocco

Version: 3 Date: 12 May 2014

Reviewer: Pattara Khamrin

Reviewer’s report:

Minor Revisions

1- Page 14, line 322: Change “sample” to “samples”
   The changing was made as indicated by the reviewer.

2- Figure 3: correct the typing of rotavirus genotypes in Figure 3(i.e “G1P8 to “G1P[8]”,
   “G2P4” to “G2P[4], etc.)
   Changes made as the reviewer indicates.

Level of interest: An article whose findings are important to those with closely related
research interests

Quality of written English: Acceptable

Statistical review: No, the manuscript does not need to be seen by a statistician.